Author's response to reviews

Title: The association between low level exposures to ambient air pollution and term low birth weight: a retrospective cohort study

Authors:

   Rose Dugandzic (rose.dugandzic@hc-sc.gc.ca)
   Linda Dodds (l.dodds@dal.ca)
   David Stieb (dave.stieb@hc-sc.gc.ca)
   Marc Smith-Doiron (marc_smith-doiron@hc-sc.gc.ca)

Version: 7  Date: 24 January 2006

Author’s response to reviews:

Figure 2 has been re-formatted.